Author's response to reviews

Title: Risk factors for biochemical recurrence after Robotic Assisted Radical Prostatectomy: a single surgeon experience

Authors:

Ryuta Tanimoto (tanimo10@gmail.com)
Yomi Fashola (yomine05@gmail.com)
Kymora B Scotland (scotland.kb@gmail.com)
Anne E Calvaresi (Anne.Calvaresi@jefferson.edu)
Leonard G Gomella (Leonard.Gomella@jefferson.edu)
Edouard J Trabulsi (Edouard.Trabulsi@jefferson.edu)
Costas D Lallas (Costas.Lallas@jefferson.edu)

Version: 4 Date: 19 March 2015

Author's response to reviews: see over
3/19/2015
Dr. Jose Karam
Section Editor
BMC Urology

Dear Dr. Karam

We would like to thank the reviewers for accepting our paper. We have carefully modified the revised manuscript accordingly. Please find below a detailed point-by-point response to each suggestion:

1. Include Tables 1 and 2 at the end of the main manuscript and remove them from the additional files.

   We have done these.

   Thank you.

Sincerely,

Ryuta Tanimoto, M.D., Ph.D.
Department of Urology
Thomas Jefferson University